

# G OPEN ACCESS

**Citation:** Kayange NM, Smart LR, Downs JA, Maskini M, Fitzgerald DW, Peck RN (2015) The Influence of HIV and Schistosomiasis on Renal Function: A Cross-sectional Study among Children at a Hospital in Tanzania. PLoS Negl Trop Dis 9(1): e0003472. doi:10.1371/journal.pntd.0003472

Editor: Zvi Bentwich, Rosetta Genomics, ISRAEL

Received: July 4, 2014

Accepted: December 12, 2014

Published: January 22, 2015

**Copyright:** © 2015 Kayange et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the paper and its Supporting Information files.

**Funding:** This project was supported in part by grants from the United States National Institute of Health/National Institute of Allergy and Infectious Diseases (Al098627), the National Institutes of Health Fogarty International Center (R25TW009337) and a United States Agency for International Development (USAID) leadership training program. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript

**RESEARCH ARTICLE** 

# The Influence of HIV and Schistosomiasis on Renal Function: A Cross-sectional Study among Children at a Hospital in Tanzania

Neema M. Kayange<sup>1</sup>\*, Luke R. Smart<sup>1,2</sup>, Jennifer A. Downs<sup>2,3</sup>, Mwanaisha Maskini<sup>1</sup>, Daniel W. Fitzgerald<sup>2,3</sup>, Robert N. Peck<sup>1,2,3</sup>

1 Department of Pediatrics, Weill Bugando School of Medicine, Catholic University of Health and Allied Sciences, Mwanza, United Republic of Tanzania, 2 Center for Global Health, Weill Cornell Medical College, New York, New York, United States of America, 3 Division of Infectious Diseases, Department of Medicine, Weill Cornell Medical College, New York, New York, United States of America

\* neemakayange@gmail.com

# Abstract

# Background

Schistosomiasis and HIV are both associated with kidney disease. Prevalence and factors associated with abnormal renal function among HIV-infected children in Africa compared to uninfected controls have not been well described in a schistosomiasis endemic area.

#### Methodology/Principal Findings

This cross-sectional study was conducted at the Sekou Toure Regional Hospital HIV clinic in Mwanza, Tanzania. A total of 122 HIV-infected children and 122 HIV-uninfected siblings were consecutively enrolled. Fresh urine was obtained for measurement of albuminuria and *Schistosoma* circulating cathodic antigen. Blood was collected for measurement of serum creatinine. Estimated glomerular filtration rate (eGFR) was calculated using the modified Schwartz equation. Renal dysfunction was defined operationally as eGFR<60mL/min/ 1.73m2 and/or albuminuria>20mg/L in a single sample. Among 122 HIV-infected children, 61/122 (50.0%) met our criteria for renal dysfunction: 54/122 (44.3%) had albuminuria>20mg/L and 9/122 (7.4%) had eGFR<60. Among 122 HIV-uninfected children, 51/122 (41.8%) met our criteria for renal dysfunction: 48/122 (39.3%) had albuminuria>20mg/L and 6/122 (4.9%) had eGFR<60. Schistosomiasis was the only factor significantly associated with renal dysfunction by multivariable logistic regression (OR = 2.51, 95% Cl 1.46–4.31, p = 0.001).

# **Conclusions/Significance**

A high prevalence of renal dysfunction exists among both HIV-infected Tanzanian children and their HIV-uninfected siblings. Schistosomiasis was strongly associated with renal dysfunction. **Competing Interests:** The authors have declared that no competing interests exist.

#### Author Summary

Ninety percent of schistosomiasis occurs in sub-Saharan Africa, where 91% of HIVinfected children reside. Both schistosomiasis and HIV affect the kidney, but their respective effects on kidney disease in children are not well described. Our prior work in HIVinfected adults demonstrated a high prevalence of kidney disease, possibly worsened by schistosomiasis, but adults are less commonly and less heavily infected with schistosomiasis than children. Therefore, we sought to describe the prevalence and factors associated with renal dysfunction (defined operationally as eGFR <60mL/min/1.73m<sup>2</sup> and/or albuminuria >20mg/L in a single urine test) among HIV-infected children and their uninfected siblings living in a schistosomiasis endemic area. We found that half of HIV-infected children and more than one third of HIV-uninfected children had renal dysfunction. Schistosomiasis was the only factor significantly associated with renal dysfunction, increasing odds of renal dysfunction by 2.5-fold. Nearly 50% of the renal dysfunction we observed in both groups could be explained by schistosomiasis. The strong association between schistosomiasis and renal dysfunction among both HIV-infected and uninfected children should remind clinicians to screen for schistosomiasis. It also ought to spur further prospective research to assess for causality and outcomes in the relationship between S. mansoni and kidney disease in children.

#### Introduction

HIV remains common in sub-Saharan Africa (SSA) where 91% of HIV-infected children reside and 1 in every 20 adults is infected [1, 2]. Kidney disease is an important complication in HIVinfected individuals and is associated with an increased risk of morbidity and mortality [3, 4]. The prevalence of kidney disease among HIV-infected adults in high-income countries ranges from 5%–50%, and is most common in patients of African descent [4]. Among more than 300 HIV-infected adults starting ART at our own hospital in Tanzania, 70% had evidence of kidney disease [5, 6].

Kidney disease among children with HIV is less well described [7]. In studies from SSA, the prevalence of markers of kidney disease among children with HIV varied greatly, ranging from 0–31.6%, depending on the methods used to evaluate the kidney [8–18]. In the few studies that included HIV-uninfected control subjects for comparison, the controls had less markers of kidney disease (0–6% versus 0–20.5%) [8, 9, 13–15]. None of these studies determined both estimated glomerular filtration rate (eGFR) and albuminuria. For this reason, the total prevalence of kidney disease among HIV-infected children in SSA remains unknown, and it is difficult to know if the kidney disease observed among children in these studies was due to HIV infection itself, medication use, or other factors that might be common among African children.

Therefore we conducted a cross-sectional study among HIV-infected children and their HIV-uninfected siblings in an area where kidney disease is common among HIV-infected adults. The objectives of our study were to determine the prevalence and correlates of renal dysfunction (defined operationally as eGFR <60mL/min/1.73m<sup>2</sup> and/or albuminuria >20mg/L in a single urine test) and to compare the prevalence of renal dysfunction among HIV-infected and HIV-uninfected children. Our hypothesis was that the prevalence of undiagnosed renal dysfunction would be 30% and 5% among HIV-infected and uninfected children, respectively, and that active schistosome infection would be associated with renal dysfunction.

#### Methods

#### Trial design and study participants

This cross-sectional study was completed between August—December 2013 in the outpatient HIV clinic at Sekou Toure Regional Hospital. Sekou Toure Hospital is located in the city of Mwanza, along the shore of Lake Victoria in northwestern Tanzania, and it serves a population of approximately 2.7 million people. The outpatient HIV clinic of Sekou Toure follows approximately 475 children who are referred to the clinic from the surrounding community-based voluntary counseling and testing centers in Mwanza. In our study, we enrolled HIV-infected children ages 2–12 years old who were attending the Sekou Toure HIV clinic. All of the mothers of these HIV-infected children were also HIV-infected, so we assume that these children were perinatally infected. The caretakers of all enrolled HIV-infected children were invited to bring uninfected siblings between the ages of 2–12 years for enrollment as controls. We tested siblings for HIV using the Determine HIV-1/2 rapid antibody test (Alere Medical Co., Ltd, Chiba, Japan) as recommended by the Tanzanian National HIV Guidelines [19]. We excluded children with fever and those for whom urine samples could not be obtained.

# Data and sample collection

At least one parent or guardian for each child was interviewed. Both HIV-infected and uninfected children were examined. A structured questionnaire was used to collect demographic information, past medical history, and clinical symptoms. Additional information collected from HIV-infected children included a history of opportunistic infections, antiretroviral therapy (ART) status, and WHO clinical stage. In both groups (HIV-infected and uninfected siblings), 4 milliliters of blood were drawn at the time of enrollment to measure the random blood glucose and serum creatinine as well as CD4<sup>+</sup> T-cell counts for HIV-infected children. Clean, midstream urine samples were collected for most children, except for a small group of children  $\leq 4$  years old in whom urine bags were used for urine collection. Urine samples were collected between 8–10AM, after  $\geq 2$  hours of fasting. Children  $\leq 4$  years old who did not urinate within 1 hour were excluded in order to minimize the inconvenience to their mothers. In order to determine the species of *Schistosoma* and intensity of infection in this population, we additionally obtained 10mL of urine and fresh fecal samples on 40 consecutive children who were CCA positive.

#### Laboratory analyses

Random blood glucose was measured using a OneTouch<sup>®</sup> glucometer (LifeScan, Inc., Milpitas, California, USA). Fresh urine samples were tested immediately for albumin using a dipstick (Micral B, Roche, Mannheim, Germany). Patients were considered to have albuminuria if the urine albumin concentration was above 20 mg/L, as per instructions provided by the manufacturer and according to our prior research [5, 6, 20], and since a concentration of >20 mg/L has been demonstrated to correlate well with elevated albumin excretion rates by standard laboratory methods [21]. A urine dipstick (Multistix 10SG, Siemens, USA) was used to test for leukocyte esterase, hematuria, nitrates, glucose and ketones. The fresh urine sample was also tested using a circulating cathodic antigen (CCA) cassette test (Rapid Medical Diagnostics, Pretoria, South Africa) to detect active *Schistosoma* infection. The CCA test indicates active schistosome infection and can be positive in the urine during infection with either species of schistosomes that are endemic in Tanzania (*S. mansoni and S. haematobium*), though its sensitivity is lower in *S. haematobium* [22–24]. CCA pointofcare testing is used widely and has been found to be more sensitive than the gold standard KatoKatz stool diagnosis of *S. mansoni*, particularly for lighter infections [25]. Following the manufacturer's instructions, any visible line in the "test" area was considered positive. Line intensities were graded as "1" (test line very faintly visible), "2" (test line visible but lighter than control line), "3" (test line equal to control line), and "4" (test line darker than control line). For fecal samples, five slides were prepared using the Kato-Katz technique, as this has been shown to have a sensitivity comparable to examining stool specimens collected on different days [26]. Ten milliliters of fresh urine was filter concentrated and examined immediately by microscopy for both contamination and for *S. haematobium*. Intensity of infection was quantified as *S. mansoni* eggs per gram of stool and *S. haematobium* eggs per 10mL of urine. The laboratory personnel who performed these analyses were blinded to the HIV-status of the study subjects.

Serum creatinine was measured using a COBAS Integra 400 Plus clinical chemistry machine (Roche, Germany), calibrated by the Creatinine Jaffe 2 Method. An estimated glomerular filtration rate (eGFR) was calculated using the modified Schwartz equation as recommended by the Kidney Disease Improving Global Outcomes (KDIGO) guidelines [27].

#### Definition

Renal dysfunction was defined operationally as an eGFR  $\leq$ 60 ml/min/1.73 m<sup>2</sup> and/or albuminuria >20 mg/L on a single urine dipstick. The severity of renal dysfunction was classified as Stage 1 (albuminuria + eGFR $\geq$ 90), Stage 2 (albuminuria + eGFR 60–89), Stage 3 (eGFR 30–59), Stage 4 (eGFR 15–29) or Stage 5 (eGFR<15). Given the age of our study population (2 to 12 years), we defined malnutrition according to the BMI-for-age child growth standards from the WHO, defining severe malnutrition as a z-score of  $\leq$ -3 [28]. We also reported weight-for-age and height-for-age for those children <5 years old.

#### Data analysis

The primary outcome was renal dysfunction as defined above. Based on the two-sample proportions Fisher's exact test, we calculated that 122 children would be needed in each group to provide >95% power (at p = 0.05) to detect the difference in prevalence of renal dysfunction between the two groups that we hypothesized (30% and 5% among HIV-infected and uninfected children respectively) while also providing >80% power to show an association between schistosomiasis and renal dysfunction if the prevalence of schistosomiasis was 50% in the children with renal dysfunction.

Data was entered into Microsoft Excel 2010 and analyzed using STATA version 12 (STATA Corporation, San Antonio, Texas). Categorical variables were described as proportions (%), and continuous variables were described as medians [interquartile range]. Univariable logistic regression analysis was performed to determine which baseline characteristics were associated with renal dysfunction. All variables significantly associated with renal dysfunction by univariable analysis were subjected to a predetermined multivariable logistic regression model, which also automatically included the variables for age, gender, and HIV status. The validity of the multivariable logistic regression model was assessed using the likelihood ratio test and by assessing for interactions. In addition, the linearity assumption was checked for continuous variables by comparing models with these variables represented continuously versus categorically. After schistosomal infection was found to be associated with renal dysfunction, we decided that we should perform an additional univariable logistic regression analysis to determine which specific markers of renal dysfunction were associated with schistosomiasis. P values of less than 0.05 were considered statistically significant.

#### Ethical issues

Ethical approval for the study was obtained from the Research and Publications Committee of Bugando Medical Centre (under whose jurisdiction Sekou Toure Regional Hospital falls), as well as the Institutional Review Board of Weill Cornell Medical College. Informed written consent was obtained from the parents and assent was obtained from children  $\geq 8$  years. The results of all tests were reported immediately to the clinician caring for the child for the sake of further management. All children with schistosomiasis were treated according to the Tanzanian National Guidelines with 40 mg/kg of praziquantel.

# Results

#### Study enrollment

During the study period, 139 HIV-infected children were seen at the Sekou Toure HIV clinic, and 17 were excluded for the following reasons: 8 were not able to provide urine samples, 5 had an acute, febrile illness, and 4 parents did not consent. In the end, 122 HIV-infected children were enrolled. For all enrolled HIV-infected children, parents were invited to bring HIV-uninfected siblings between the ages of 2–12 years for enrollment as controls. A total of 132 siblings were screened, and 10 were excluded for the following reasons: 6 were not able to provide urine samples, and 4 had an acute, febrile illness. A total of 122 siblings were enrolled. The characteristics of excluded children did not differ between the 2 groups.

# **Baseline characteristics**

<u>Table 1</u> describes the baseline characteristics of the 2 groups. Among the 122 HIV-infected children, the median age was 8 years [4-11] and 63/122 (51.6%) were female. Among the 122 HIV-uninfected children, the median age was 8 years [5-10], and 55/122 (45.1%) were female.

Historical factors that were significantly different between the 2 groups included cough (26/122 [21.3%] vs. 12/122 [9.8%], p = 0.01), history of tuberculosis (10/122 [8.2%] vs. 1/122 [0.8%], p = 0.005), recurrent pneumonia (11/122 [9.0%] vs. 3/122 [2.5%], p = 0.03), and papular pruritic eruptions (12/122 [9.8%] vs. 2/122 [1.6%], p = 0.01). Physical exam factors that were significantly different between the groups were pallor (10/122 [8.2%] vs. 2/122 [1.6%], p = 0.02), thrush (4/122 [3.3%] vs. 0/122 [0%], p = 0.04) and lymphadenopathy (23/122 [18.9%] vs. 3/122 [2.5%] p=<0.0001). No children reported history of neurologic disease and none had signs of neurologic disease on physical examination. The BMI-for-age z-scores were similar in the 2 groups. For the children <5 years old, the weight-for-age and height-for-age z-scores were also similar in the 2 groups (-0.76 [-1.57-0.05] vs. -0.48 [-1.12-0.21], p = 0.47 and -1.58 [-2.16-0.13] vs. -1.33 [-2.54-0.49], p = 0.47 respectively).

# Renal dysfunction outcomes

<u>Table 2</u> describes the renal dysfunction outcomes (defined operationally as eGFR <60mL/min/  $1.73m^2$  and/or albuminuria >20mg/L in a single urine test). Among 122 HIV-infected children, 61/122 (50.0%) met our criteria for renal dysfunction: 54 (44.3%) had albuminuria, and 9 (7.4%) had an eGFR <60. Among 122 HIV-uninfected children, 51 (41.8%) met our criteria for renal dysfunction: 48 (39.3%) had albuminuria and 6 (4.9%) had an eGFR < 60.

# Factors associated with renal dysfunction

<u>Table 3</u> shows the results of the univariable analysis for factors associated with renal dysfunction. In the univariable analysis only the presence of schistosomiasis was significantly associated with renal dysfunction (OR = 2.51, 95%CI 1.46-4.31, p = 0.001). Higher intensity of

#### Table 1. Baseline characteristics of HIV-infected children and HIV-uninfected siblings.

	HIV-infected cases (n = 122) n (%) or median [IQR]	HIV-uninfected siblings (n = 122) n (%) or median [IQR]	p-value
Age (years)	8 [4–11]	8 [5–10]	0.80
Sex			
Male	59 (48.4)	67 (54.9)	0.31
Female	63 (51.6)	55 (45.1)	
Symptoms			
Rash	12 (9.8)	6 (4.9)	0.14
Oral Sore	2 (1.6)	1 (0.8)	0.56
Diarrhea	1 (0.8)	1 (0.8)	1.0
Cough	26 (21.3)	12 (9.8)	0.01
Edema	0 (-)	0 (-)	-
Past Medical History			
Tuberculosis	10 (8.2)	1 (0.8)	0.005
Herpes Zoster	3 (2.5)	0 (-)	0.08
Recurrent Pneumonia	11 (9.0)	3 (2.5)	0.03
Oral Thrush	9 (7.4)	4 (3.3)	15
Papular Pruritic Eruptions	12 (9.8)	2 (1.6)	0.01
Known Kidney Disease	0 (-)	0 (-)	-
Co-trimoxazole Use	58 (52.3)	NA	-
Physical Examination			
Pallor	10 (8.2)	2 (1.6)	0.02
Thrush	4 (3.3)	0 (-)	0.04
Lymphadenopathy	23 (18.9)	3 (2.5)	<0.001
CD4 Count (cells/uL)	521.5 [302–828]	NA	-
Malnutrition			
BMI-for-age Z-scores	-0.17 [-1.01–1.26]	0.17 [-0.67–0.99]	0.67
Severe malnutrition*	1 (0.8)	2 (1.6)	0.57
WHO Clinical Staging			
1	3 (2.5)	NA	-
2	39 (32.2)	NA	-
3	67 (55.4)	NA	-
4	12 (9.9)	NA	-
Random Blood Glucose (mmol/L)	5.8 [4.9–6.7]	6 [5.2–6.8]	0.61

\*Severe malnutrition has been defined as a BMI-for-age Z score  $\leq$ -3.

doi:10.1371/journal.pntd.0003472.t001

schistosomiasis was associated with higher prevalence of renal dysfunction (OR = 1.3 (1+) vs. OR $\approx$ 4 (2+/3+), p for trend = 0.001). By multivariable logistic regression analysis including schistosomiasis, age, sex and HIV status, schistosomiasis remained the only factor associated with renal dysfunction (OR = 2.40, 95%CI 1.37–4.17, p = 0.002). HIV infection was not significantly associated with renal dysfunction by either univariable or multivariable analysis. There was more renal dysfunction among children with higher WHO clinical stage, but this relationship was not statistically significant (p = 0.23). Of note, though, the association between schistosomiasis and renal dysfunction was somewhat stronger among HIV-infected children than their HIV-uninfected siblings (OR = 3.04, 95%CI 1.39–6.68, p = 0.005 vs. OR = 2.08, 95%CI 0.97–4.46, p = 0.06).

	HIV-infected cases (n = 122) n (%) or median [IQR]	HIV-uninfected siblings (n = 122) n (%) or median [IQR]	p-value
Renal dysfunction*	61 (50.0)	51 (41.8)	0.20
eGFR <sup>†</sup> (ml/min/1.73 m <sup>2</sup> )	112.9 [84.9–143–6]	113.3 [91.4–134.2]	1.0
eGFR Category			
Stage 1 ( $\geq$ 90)	88 (72.1)	95 (77.9)	0.43
Stage 2 (60–89)	25 (20.5)	21 (17.2)	
Stage 3 (30–59)	9 (7.4)	5 (4.1)	
Stage 4 (15–29)	0 (-)	0 (-)	
Stage 5 (< 15)	0 (-)	1 (0.8)	
Albuminuria			
Negative	68 (55.7)	74 (60.7)	0.88
>20 mg/L	38 (31.2)	35 (28.7)	
>50 mg/L	11 (9.0)	9 (7.4)	
>100 mg/L	5 (4.1)	4 (3.3)	
Hematuria	12 (9.8)	13 (10.6)	0.83
Pyuria	8 (6.6)	6 (4.9)	0.58

Table 2. Markers of renal dysfunction in HIV-infected children and HIV-uninfected siblings.

\*the primary study outcome (defined operationally as eGFR <60mL/min/1.73m<sup>2</sup> and/or albuminuria >20mg/L in a single urine test) <sup>†</sup>eGFR = estimated glomerular filtration rate by modified Schwartz equation

doi:10.1371/journal.pntd.0003472.t002

#### Renal dysfunction and Schistosoma infection

<u>Table 4</u> compares the prevalence of renal dysfunction among *Schistosoma* infected children and *Schistosoma* uninfected children regardless of HIV status. An eGFR less than 60 ml/min/  $1.73m^2$  was more common in the *Schistosoma* infected children (7.2% versus 5.6%, OR 2.61, 95% CI 1.35–5.04, p = 0.01). There was also a higher prevalence of albuminuria among *Schistosoma* infected children (55.9% versus 34.4%, OR 2.92, 95% CI 1.61–5.27, p = 0.001). The overall prevalence of renal dysfunction was 60.7% among *Schistosoma* infected children and 38.1% among uninfected children (OR 2.51, 95% CI 1.46–4.31, p = 0.001).

#### Schistosome species

None of the 40 subjects who provided urine for microscopy had eggs of *S. haematobium* detectable by microscopy. Of the 7 subjects from whom stool samples were available, 6/7 (85.7%) had *S. mansoni* eggs detected, with concentrations ranging from 17 to 75 eggs per gram.

#### Discussion

In our study, half of the HIV-infected children attending an HIV clinic in the Lake Zone of northwestern Tanzania had evidence of renal dysfunction (defined operationally as eGFR <60mL/min/1.73m<sup>2</sup> and/or albuminuria >20mg/L in a single urine test): 44.3% had albuminuria >20mg/L and 7.4% had an eGFR <60 ml/min/1.73 m<sup>2</sup>. These rates are higher than those found in other studies from SSA. Four other countries in SSA (Burkina Faso, Democratic Republic of Congo, Nigeria, and Zimbabwe) have reported the prevalence of markers of kidney disease among HIV-infected children as ranging from 0–31.6% using methodologies similar to ours [8–18]. The reasons for the higher prevalence of renal dysfunction among HIV-infected children in the Lake Zone compared to prior studies in SSA is not known, but our findings are consistent with the findings among HIV-infected adults in our region [5, 6].

	With Renal Dysfunction (n = 112) n (%) or median [IQR]	Without Renal Dysfunction (n = 132) n (%) or median [IQR]	Odds Ratio (95% CI)	p-value
Age (years)	9 [5–11]	8 [4.5–10]	1.50	0.22
Sex				
Male	57 (50.89)	61 (46.21)	0.83 (0.51–1.37	0.47
Female	55 (49.11)	71 (53.79)		
Symptoms				
Rash	5 (4.46)	13 (9.85)	0.43 (0.15–1.24)	0.12
Oral Sore	1 (0.89)	2 (1.52)	0.59 (0.05-6.54)	0.66
Diarrhea	1 (0.89)	1 (0.76)	1.18 (0.07–19.09)	0.91
Cough	17 (15.18)	21 (15.91)	0.95 (0.47-1.90)	0.88
Edema	0 (-)	0 (-)	-	-
Past Medical History				
Fever	5 (4.46)	9 (6.82)	0.64 (0.21–1.96)	0.43
тв	8 (7.14)	3 (2.27)	3.30 (0.86–12.78)	0.08
Herpes Zoster	1 (0.89)	3 (2.27)	1.18 (0.07–19.09)	0.91
Recurrent Pneumonia	4 (3.57)	2 (1.52)	2.41 (0.43–13.40)	0.32
Oral Thrush	0 (-)	3 (2.27)	-	-
Co-trimoxazole use	33 (29.46)	33 (25.00)	1.25 (0.71–2.21)	0.43
Physical Examination				
Pallor	5 (4.5)	7 (5.3)	0.83 (0.26–2.71)	0.76
Thrush	1 (0.9)	3 (2.3)	0.39 (0.04–3.78)	0.41
Lymphadenopathy	16 (14.3)	10 (7.6)	2.03 (0.88-4.68)	0.10
Severe Malnutrition	1 (0.9)	2 (1.5)	0.59 (0.05-6.54)	0.66
CD4 count (cells/uL)*				
Mean	535.8 (399.3)	644.7 (363.4)	1.58	0.11
Median	481 [204–747]	612 [341–895)	0.82	0.37
WHO Clinical Stage*				
1	0 (-)	3 (4.9)	-	
2	16 (26.7)	23 (37.7)	0.35 (0.09–1.35)	
3	36 (60.0)	31 (50.8)	0.58 (0.16-2.12)	0.23
4	8 (13.3)	4 (6.6)	1	
Random Blood Glucose	6.2 [5.2–6.8]	5.8 [5-6.7]	2.25	0.13
Schistosoma positive <sup>†</sup>	51 (45.5)	33 (25.00)	2.51 (1.46-4.31)	0.001
Schistosoma intensity <sup>†</sup>			· · · ·	
Negative	61 (54.5)	99 (75.0)	1	
1+	12 (10.7)	15 (11.4)	1.30 (0.57–2.96)	
2+	13 (11.6)	5 (3.8)	4.22 (1.43–12.4)	0.001
3+	26 (23.2)	13 (9.9)	3.25 (1.55–6.79)	
HIV-infected	61 (54.5)	61 (46.2)	1.39 (0.84–2.31)	0.19

Table 3. Factors associated with renal dysfunction among HIV-infected children and HIV-uninfected siblings by univariable logistic regression.

\*For the 122 HIV-infected cases only

<sup>†</sup>Schistosoma infection and intensity were both determined with the urine CAA assay.

doi:10.1371/journal.pntd.0003472.t003

Surprisingly, the prevalence of renal dysfunction was equally high among HIV-uninfected siblings. More than one-third of HIV-uninfected siblings had evidence of renal dysfunction: 39.3% had albuminuria >20 mg/L and 4.9% had eGFR<60 ml/min/1.73 m<sup>2</sup>. In other studies among children in SSA that have included HIV-uninfected controls, the prevalence of markers

	<i>Schistosoma</i> +ve (n = 84) Number (%) or Median [IQR]	<i>Schistosoma</i> – ve (n = 160) Number (%) or Median [IQR]	Odds ratio (95%Cl)	p-value
Renal dysfunction*	51 (60.7)	61 (38.1)	2.51 (1.46–4.31)	0.001
eGFR <sup>†</sup> (ml/min/1.73m <sup>2</sup> )	102.7 [77.5–131.7]	115.4 [92.9–141.8]	3.6	0.059
eGFR <60ml/min/1.73m <sup>2</sup>	6 (7.2)	9 (5.6)	2.61 (1.35–5.04)	0.014
Albuminuria	47 (55.9)	55 (34.4)	2.92 (1.61–5.27)	0.001
Hematuria	14 (16.7)	11 (6.9)	2.71 (1.17–6.27)	0.02
Pyuria	8 (4.9)	6 (7.1)	2.70 (0.91–8.06)	0.07

Table 4. Markers of renal dysfunction associated with Schistosoma infection by univariable logistic regression.

\*the primary study outcome (defined operationally as eGFR <60mL/min/1.73m<sup>2</sup> and/or albuminuria >20mg/L in a single urine test) <sup>†</sup>eGFR = estimated glomerular filtration rate as calculated by modified Schwartz equation

doi:10.1371/journal.pntd.0003472.t004

of kidney disease in the control group has been low (0-6%) [8, 9, 13–15]. The higher prevalence of renal dysfunction among HIV-uninfected children in our study is likely related to the unique nature of our control group. In order to minimize differences in socioeconomic factors, household exposures and genetics, we chose HIV-uninfected siblings as our control group, whereas prior studies have all used children from the pediatric outpatient clinics of the hospital in which the study was conducted as their controls [8, 9, 13-15]. The high prevalence of renal dysfunction that we observed among HIV-uninfected controls could therefore be related to either in utero HIV exposure or a high population prevalence of renal dysfunction. Because the dates of maternal HIV infection and viral suppression were not known, we could not confirm the HIV-exposure status among HIV-uninfected control siblings, but we suspect that most if not all were exposed since they were born within ~5 years of their HIV-infected sibling. HIV exposure, even in the absence of infection, has been associated with multiple abnormalities that may affect the development and function of the kidney in childhood [29-31]. On the other hand, the high prevalence of renal dysfunction among controls could also reflect a high community prevalence of renal dysfunction among children, possibly related to the known high prevalence of schistosomiasis. In order to investigate this possibility, we are currently planning a longitudinal study to examine a cohort of HIV-uninfected, unexposed school children in the Lake Zone.

Schistosomiasis was strongly associated with renal dysfunction among HIV-infected and uninfected children (OR = 2.51, 95% CI 1.46–4.31), and a higher intensity schistosome infection was associated with even more renal dysfunction (p = 0.001 for trend). Approximately 46% of children with renal dysfunction had schistosomiasis by CCA testing compared to 25% of children without renal dysfunction. Multiple studies and epidemiological evidence have shown that *S. haematobium* infection may cause kidney disease (particularly albuminuria) in both adults and children [32-34], but all of the 40 consecutive urine samples we tested were negative for *S. haematobium*, and <20% of CCA-positive subjects had the hematuria that is typical for this infection. The dominant species of schistosomiasis in our region, *S. mansoni* [35, 36], is also known to cause glomerulonephritis and kidney disease [37]. The prevalence of proteinuria among adult subjects infected with *S. mansoni* has been reported to be as high as 20% in Egypt [38, 39] and 15% in Brazil [40, 41], and the severity and irreversibility of the disease has been demonstrated in several clinicopathologic and experimental studies [42-44]. The association between markers of kidney disease and *S. mansoni* in children, by contrast, has only been investigated in two small studies in SSA [45, 46] which did not find an association

between *S. mansoni* and overt proteinuria, but neither one of them investigated both eGFR and albuminuria as we have done. If further studies confirm the relationship between *S. mansoni* infection and kidney disease among children, schistosomiasis may become an important target for prevention of kidney disease in our population.

Although schistosomiasis could explain a large proportion of the renal dysfunction in our study, 55% of children with renal dysfunction did not have evidence of current schistosomiasis. In these subjects multiple other factors may be contributing to their renal dysfunction such as acute glomerulonephritis, infections (malaria, recurrent diarrhea), or genetic factors. The APOL1 gene mutation, for example, is known to be associated with renal disease in African populations [47], younger age of onset of kidney disease [48] and faster decline in kidney function [49]. Further studies are needed to determine the factors other than schistosomiasis that might be contributing to the high prevalence of renal dysfunction that we observed. Kidney disease occurring at a young age may lead to end stage renal disease or other complications (e.g. hypertension) in adulthood; therefore targeted screening for early detection of kidney disease and treatment of reversible factors are high priorities. Many simple diagnostic tools such as light microscopy of the urine sediment are currently underutilized in screening efforts. In addition, the implementation of proven strategies for prevention and treatment, such as early antimicrobial therapy for severe infections and rapid correction of hypovolemic shock, must be accelerated.

Our study has several limitations. First and foremost, this was a cross-sectional study and, therefore, neither confirmation of chronic kidney disease nor causality in the relationship between schistosomiasis and kidney disease could be examined. In addition, several gold standard investigations for HIV and kidney disease, such as quantitative HIV viral load testing, urine albumin-to-creatinine ratio, and kidney biopsies were not performed since they were not available in our region at the time of the study. Our operational definition of renal dysfunction may have resulted in information bias, with over diagnosis of kidney disease, and we are currently planning to assess our findings with a study using the standard KDIGO definition of chronic kidney disease. Also, the exclusion of children who could not produce urine samples, as well as the possibility that HIV-infected patients may have received previous praziquantel therapy could have caused some selection bias and underestimation of the population prevalences of renal dysfunction and schistosomiasis, respectively.

In conclusion, our study identified a high prevalence of renal dysfunction (defined operationally as eGFR <60mL/min/1.73m<sup>2</sup> and/or albuminuria >20mg/L in a single urine test) among HIV-infected Tanzanian children attending our pediatric HIV clinics. Almost 50% of both HIV-infected children and their siblings had renal dysfunction, and 6% had an eGFR <60 mL/min/1.73 m<sup>2</sup>. Surprisingly, the prevalence of renal dysfunction among HIV-uninfected siblings was similar to the HIV-infected children. This may be related to either in utero HIV exposure or a high community-wide prevalence of renal dysfunction in children, and further studies are urgently needed to distinguish these possibilities. Schistosomiasis was strongly associated renal dysfunction in this population, and the predominant species of schistosomes in our region is *S. mansoni* (not *S. haematobium*). Schistosomiasis may be an important target for prevention of kidney disease in children in sub-Saharan Africa.

#### **Supporting Information**

**S1 Checklist. Completed STROBE checklist for cross-sectional studies.** (DOCX)

#### **Author Contributions**

Conceived and designed the experiments: NMK RNP DWF. Performed the experiments: NMK MM. Analyzed the data: NMK RNP LRS JAD. Contributed reagents/materials/analysis tools: DWF JAD RNP NMK. Wrote the paper: NMK LRS MM JAD RNP DWF.

#### References

- 1. World Health Organization (n.d.) HIV/AIDS. WHO Fact Sheet 360. <u>www.who.int/mediacentre/</u> factsheets/fs360/en/.
- 2. United Nations Children's Fund (2013) Towards an AIDS-Free Generation—Children and AIDS: Sixth Stocktaking Report. UNICEF.
- 3. Choi Al, O'Hare A, Rodriguez R (2007) Update on HIV-associated Nephropathy. Nephrol Rounds 5.
- Fabian J, Naicker S (2009) HIV and kidney disease in sub-Saharan Africa. Nat Rev Nephrol 5: 591–598. doi: <u>10.1038/nrneph.2009.141</u> PMID: <u>19776781</u>
- Msango L, Downs JA, Kalluvya SE, Kidenya BR, Kabangila R, et al. (2011) Renal dysfunction among HIV-infected patients starting antiretroviral therapy. AIDS 25: 1421–1425. doi: <u>10.1097/QAD.</u> <u>0b013e328348a4b1</u> PMID: <u>21572304</u>
- Mpondo BCT, Kalluvya SE, Peck RN, Kabangila R, Kidenya BR, et al. (2014) Impact of Antiretroviral Therapy on Renal Function among HIV-Infected Tanzanian Adults: A Retrospective Cohort Study. PLoS One 9: e89573. doi: <u>10.1371/journal.pone.0089573</u> PMID: <u>24586882</u>
- 7. Bhimma R, Purswani MU, Kala U (2013) Kidney disease in children and adolescents with perinatal HIV-1 infection. J Int AIDS Soc 16: 18596. doi: <u>10.7448/IAS.16.1.18596</u> PMID: <u>23782479</u>
- Abiodun MT, Iduoriyekemwen NJ, Abiodun PO (2012) Cystatin C-Based Evaluation of Kidney Function of HIV-Infected Children in Benin City, Southern Nigeria. Int J Nephrol 2012: 861296. doi: <u>10.1155/</u> 2012/861296 PMID: 23213527
- Bertilla Uzoma E, Henrietta Uchenna O, Anthony Nnaemeka I, Tagbo O (2012) Screening for Microalbuminuria in HIV-Positive Children in Enugu. Int J Nephrol 2012: 805834. doi: <u>10.1155/2012/805834</u> PMID: 22830018
- Coulibaly G, Kouéta F, Ouédraogo O, Dao L, Lengani A, et al. (2013) Prevalence of proteinuria in children followed-up for HIV infection at Pediatric University Hospital Charles-de-Gaulle (CHUP-CDG) of Ouagadougou. Bull Soc Pathol Exot 106: 13–17. doi: <u>10.1007/s13149-012-0270-9</u> PMID: <u>23315307</u>
- Dondo V, Mujuru H a, Nathoo KJ, Chirehwa M, Mufandaedza Z (2013) Renal abnormalities among HIV-infected, antiretroviral naive children, Harare, Zimbabwe: a cross-sectional study. BMC Pediatr 13: 75. doi: 10.1186/1471-2431-13-75 PMID: 23663553
- Eke FU, Anochie IC, Okpere AN, Eneh AU, Ugwu RO, et al. (2010) Microalbuminuria in children with human immunodeficiency virus (HIV) infection in Port Harcourt, Nigeria. Niger J Med 19: 298–301. doi: <u>10.4314/njm.v19i3.60214</u> PMID: <u>20845635</u>
- **13.** Ekulu PM, Nseka NM, Aloni MN, Gini J-LE, Makulo J-R, et al. (2012) Prevalence of proteinuria and its association with HIV/AIDS in Congolese children living in Kinshasa, Democratic Republic of Congo. Néphrologie & thérapeutique 8: 163–167. doi:10.1016/j.nephro.2011.09.004.
- Esezobor CI, Iroha E, Oladipo O, Onifade E, Soriyan OO, et al. (2010) Kidney function of HIV-infected children in Lagos, Nigeria: using Filler's serum cystatin C-based formula. J Int AIDS Soc 13: 17. doi: <u>10.1186/1758-2652-13-17</u> PMID: <u>20482807</u>
- Esezobor CI, Iroha E, Onifade E, Akinsulie AO, Temiye EO, et al. (2010) Prevalence of proteinuria among HIV-infected children attending a tertiary hospital in Lagos, Nigeria. J Trop Pediatr 56: 187– 190. doi: <u>10.1093/tropej/fmp090</u> PMID: <u>19793893</u>
- Iduoriyekemwen NJ, Sadoh WE, Sadoh AE (2013) Prevalence of renal disease in Nigerian children infected with the human immunodeficiency virus and on highly active anti-retroviral therapy. Saudi J Kidney Dis Transpl 24: 172–177. doi: <u>10.4103/1319-2442.106364</u> PMID: <u>23354220</u>
- Ikpeme EE, Ekrikpo UE, Akpan MU, Ekaidem SI (2012) Determining the prevalence of human immunodeficiency virus-associated nephropathy (HIVAN) using proteinuria and ultrasound findings in a Nigerian paediatric HIV population. Pan Afr Med J 11: 13. PMID: <u>22368756</u>
- Fredrick F, Ruggajo P, Maro EE, Iversen BM, Basu G (2012) Renal manifestations and associated factors among HIV infected children at Muhimbili National Hospital, Dar es Salaam, Tanzania. BMC Infect Dis 12: O11. doi: 10.1186/1471-2334-12-S1-O11
- National AIDS Control Programme (NACP) Tanzania (2012) National Guidelines for the Management of HIV and AIDS. 4th ed. Dar es Salaam, Tanzania: The United Republic of Tanzania Ministry of Health and Social Welfare Tanzania Mainland.

- Janmohamed MN, Kalluvya SE, Mueller A, Kabangila R, Smart LR, et al. (2013) Prevalence of chronic kidney disease in diabetic adult out-patients in Tanzania. BMC Nephrol 14: 183. doi: <u>10.1186/1471-</u> 2369-14-183 PMID: 24228774
- Hasslacher C (1993) Clinical significance of microalbuminuria and evaluation of the Micral-Test. Clin Biochem 26: 283–287. doi: 10.1016/0009-9120(93)90126-Q PMID: 8242889
- 22. Obeng BB, Aryeetey YA, de Dood CJ, Amoah AS, Larbi IA, et al. (2008) Application of a circulatingcathodic-antigen (CCA) strip test and real-time PCR, in comparison with microscopy, for the detection of Schistosoma haematobium in urine samples from Ghana. Ann Trop Med Parasitol 102: 625–633. doi: 10.1179/136485908X337490 PMID: 18817603
- Stothard JR, Sousa-Figueiredo JC, Standley C, Van Dam GJ, Knopp S, et al. (2009) An evaluation of urine-CCA strip test and fingerprick blood SEA-ELISA for detection of urinary schistosomiasis in schoolchildren in Zanzibar. Acta Trop 111: 64–70. doi: <u>10.1016/j.actatropica.2009.02.009</u> PMID: <u>19426665</u>
- Ashton RA, Stewart BT, Petty N, Lado M, Finn T, et al. (2011) Accuracy of circulating cathodic antigen tests for rapid mapping of Schistosoma mansoni and S. haematobium infections in Southern Sudan. Trop Med Int Health 16: 1099–1103. doi: 10.1111/j.1365-3156.2011.02815.x
- Colley DG, Binder S, Campbell C, King CH, Tchuem Tchuenté L-A, et al. (2013) A five-country evaluation of a point-of-care circulating cathodic antigen urine assay for the prevalence of Schistosoma mansoni. Am J Trop Med Hyg 88: 426–432. doi: <u>10.4269/ajtmh.12-0639</u> PMID: <u>23339198</u>
- Berhe N, Medhin G, Erko B, Smith T, Gedamu S, et al. (2004) Variations in helminth faecal egg counts in Kato-Katz thick smears and their implications in assessing infection status with Schistosoma mansoni. Acta Trop 92: 205–212. doi: 10.1016/j.actatropica.2004.06.011 PMID: 15533288
- Kidney Disease: Improving Global Outcomes (KDIGO) CKD Work Group (2013) KDIGO 2012 Clinical Practice Guidelines for the Evaluation and Management of Chronic Kidney Disease. Kidney Int Suppl 3: 1–150. doi:10.1038/kisup.2012.76.
- 28. World Health Organization (n.d.) Child Growth Standards: BMI-for-age. Available: <u>http://www.who.int/</u> childgrowth/standards/bmi\_for\_age/en/.
- Sugandhi N, Rodrigues J, Kim M, Ahmed S, Amzel A, et al. (2013) HIV-exposed infants: rethinking care for a lifelong condition. AIDS 27 Suppl 2: S187–95. doi: <u>10.1097/QAD.000000000000090</u> PMID: <u>24361628</u>
- 30. Moraleda C, de Deus N, Serna-Bolea C, Renom M, Quintó L, et al. (2014) Impact of HIV exposure on health outcomes in HIV-negative infants born to HIV-positive mothers in Sub-Saharan Africa. J Acquir Immune Defic Syndr 65: 182–189. doi: 10.1097/QAI.000000000000019 PMID: 24442224
- Owor M, Mwatha A, Donnell D, Musoke P, Mmiro F, et al. (2013) Long-term follow-up of children in the HIVNET 012 perinatal HIV prevention trial: five-year growth and survival. J Acquir Immune Defic Syndr 64: 464–471. doi: 10.1097/QAI.00000000000015 PMID: 24121753
- King CH, Lombardi G, Lombardi C, Greenblatt R, Hodder S, et al. (1988) Chemotherapy-based control of schistosomiasis haematobia. I. Metrifonate versus praziquantel in control of intensity and prevalence of infection. Am J Trop Med Hyg 39: 295–305.
- King CH, Lombardi G, Lombardi C, Greenblatt R, Hodder S, et al. (1990) Chemotherapy-based control of schistosomiasis haematobia. II. Metrifonate vs. praziquantel in control of infection-associated morbidity. Am J Trop Med Hyg 42: 587–595.
- Ezzat E, Osman RA, Ahmet KY, Soothill JF (1974) The association between Schistosoma haematobium infection and heavy proteinuria. Trans R Soc Trop Med Hyg 68: 315–318. doi: <u>10.1016/0035-9203(74)90040-6</u> PMID: <u>4417931</u>
- Downs JA, van Dam GJ, Changalucha JM, Corstjens PLAM, Peck RN, et al. (2012) Association of Schistosomiasis and HIV infection in Tanzania. Am J Trop Med Hyg 87: 868–873. doi: <u>10.4269/ajtmh.</u> <u>2012.12-0395</u> PMID: <u>23033399</u>
- Efraim L, Peck RN, Kalluvya SE, Kabangila R, Mazigo HD, et al. (2013) Schistosomiasis and impaired response to antiretroviral therapy among HIV-infected patients in Tanzania. J Acquir Immune Defic Syndr 62: e153–6. doi: 10.1097/QAI.0b013e318282a1a4 PMID: 23760064
- Andrade ZA, Andrade SG, Sadigursky M (1971) Renal changes in patients with hepatosplenic schistosomiasis. Am J Trop Med Hyg 20: 77–83. PMID: <u>5567743</u>
- Barsoum R, Abdel-Raham Y, Francis M, El-Kaliouby A, Zakareya S, et al. (1992) Patterns of glomerular injury associated with hepato-intestinal schistosomiasis. Proc XII Egypt Cong Nephrol, Cairo.
- Sobh M, Moustafa F, El-Arbagy A, El-Din MS, Shamaa S, et al. (1990) Nephropathy in asymptomatic patients with active Schistosoma mansoni infection. Int Urol Nephrol 22: 37–43. doi: <u>10.1007/</u> BF02550434 PMID: 2116378

- 40. Bina J, Andrade Z, Dietze R, Prata A (1985) A field study of proteinuria in individuals infected with Schistosoma mansoni. Rev Soc Bras Med Trop 18: 7–10. doi: <u>10.1590/S0037-86821985000100003</u>
- Rocha H, Cruz T, Brito E, Susin M (1976) Renal involvement in patients with hepatosplenic Schistosomiasis mansoni. Am J Trop Med Hyg 25: 108–115. PMID: 1259074
- Martinelli R, Pereira LJ, Brito E, Rocha H (1995) Clinical course of focal segmental glomerulosclerosis associated with hepatosplenic schistosomiasis mansoni. Nephron 69: 131–134. doi: <u>10.1159/</u> 000188427 PMID: 7723893
- Sobh MA, Moustafa FE, Sally SM, Deelder AM, Ghoniem MA (1988) Effect of anti-schistosomal treatment on schistosomal-specific nephropathy. Nephrol Dial Transplant 3: 744–751. PMID: <u>3147416</u>
- Sobh MA, Moustafa FE, El-Housseini F, Basta MT, Deelder AM, et al. (1987) Schistosomal specific nephropathy leading to end-stage renal failure. Kidney Int 31: 1006–1011. doi: <u>10.1038/ki.1987.99</u> PMID: <u>3108566</u>
- Elsheikh M, Doehring-Schwerdtfeger E, Kaiser C, Abdelrahim IM, Ali GM, et al. (1989) Renal function in Sudanese school children with Schistosoma mansoni infection. Pediatr Nephrol 3: 259–264. doi: <u>10.</u> <u>1007/BF00858526</u> PMID: <u>2518450</u>
- 46. Johansen M V, Simonsen PE, Butterworth AE, Ouma JH, Mbugua GG, et al. (1994) A survey of Schistosoma mansoni induced kidney disease in children in an endemic area of Machakos District, Kenya. Acta Trop 58: 21–28. doi: 10.1016/0001-706X(94)90118-X PMID: 7863851
- 47. Tayo BO, Kramer H, Salako BL, Gottesman O, McKenzie CA, et al. (2013) Genetic variation in APOL1 and MYH9 genes is associated with chronic kidney disease among Nigerians. Int Urol Nephrol 45: 485–494. doi: 10.1007/s11255-012-0263-4 PMID: 22956460
- Kanji Z, Powe CE, Wenger JB, Huang C, Ankers E, et al. (2011) Genetic variation in APOL1 associates with younger age at hemodialysis initiation. J Am Soc Nephrol 22: 2091–2097. doi: <u>10.1681/ASN.</u> 2010121234 PMID: 21997398
- 49. Parsa A, Kao WHL, Xie D, Astor BC, Li M, et al. (2013) APOL1 risk variants, race, and progression of chronic kidney disease. N Engl J Med 369: 2183–2196. doi: <u>10.1056/NEJMoa1310345</u> PMID: <u>24206458</u>